Author’s response to reviews

Title: Tissue-Infiltrating Lymphocytes Signature Predicts Survival in Patients with Early/Intermediate Stage Hepatocellular Carcinoma

Authors:

yinghong shi (shi.yinghong@zs-hospital.sh.cn)
Meng-Xin Tian MD (tianmengxingyu@163.com)
Wei-Ren Liu MD (liu.weiren@zs-hospital.sh.cn)
Han Wang (15211210028@fudan.edu.cn)
Yu-Fu Zhou MD (maikestar99@163.com)
Lei Jin (JLfudan2013@163.com)
Xi-Fei Jiang (14211210029@fudan.edu.cn)
Chen-Yang Tao (363392117@qq.com)
Zheng Tang (897594561@qq.com)
Pei-Yun Zhou (15111210068@fudan.edu.cn)
Yuan Fang (17317822004@163.com)
Wei-Feng Qu (527544659@qq.com)
Zhen-Bin Ding MD (ding.zhenbin@zs-hospital.sh.cn)
Yuan-Fei Peng MD (peng.yuanfei@zs-hospital.sh.cn)
Zhi Dai (dai.zhi@zs-hospital.sh.cn)
Shuang-Jian Qiu MD (qiu.shuangjian@zs-hospital.sh.cn)
Jian Zhou (zhou.jian@zs-hospital.sh.cn)
Wan Yee Lau (josephlau@cuhk.edu.hk)
Jia Fan (fan.jia@zs-hospital.sh.cn)

Version: 4 Date: 29 Apr 2019
Author's response to reviews:

Dear Alessandro Recchioni, We have revised the section of "Availability of data and materials" as suggested. The sentence were as follows: 'The data that support the findings of this study are available from the department of liver surgery in Zhongshan hospital but restrictions apply to the availability of these data, which were used under license for the current study, and so are not publicly available. Data are however available from the authors upon reasonable request and with permission of the department of liver surgery in Zhongshan hospital.' Best wishes!